Author's response to reviews

Title: An evaluation of a data linkage training workshop for research ethics committees

Authors:

Kate M Tan (kate.tan@telethonkids.org.au)
Felicity S Flack (felicity.flack@telethonkids.org.au)
Natasha L Bear (natasha.bear@telethonkids.org.au)
Judy A Allen (judy.allen@uwa.edu.au)

Version: 4 Date: 29 January 2015

Author's response to reviews: see over
29 January 2015

To whom it may concern,

Thank you for your email notifying us that the manuscript entitled “An evaluation of a data linkage training workshop for research ethics committees” has been accepted. In response to the additional Editorial request, please note that information about whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian has been added to the methods section as suggested (L164-166). Please find attached a revised copy of the manuscript.

Yours sincerely,

Kate Tan